Author's response to reviews

Title: Annular leucocytoclastic vasculitis associated with anti-tuberculous medications: a case report

Authors:

Wanjarus Roongpisuthipong (rr_wanjarus@hotmail.com)
Kunlawat Thadanipon (kunlawat@gmail.com)
Kumutnart Chanprapaph (rakcp@mahidol.ac.th)

Version: 2 Date: 22 July 2012

Author's response to reviews:

We believe it is the first report of annular leucocytoclastic vasculitis associated with anti-tuberculous medications. It is an original case report of interest to a particular dermatologist or will it have a boarder clinical impact across more than one area of medicine.